Author's response to reviews

Title: Cause-of-death ascertainment for deaths that occur outside hospitals in Thailand: application of verbal autopsy methods

Authors:

Warangkana Polprasert (pwarang_p@yahoo.com)
Chalapati Rao (c.rao@sph.uq.edu.au)
Timothy Adair (t.adair@uq.edu.au)
Junya Pattaraarchachai (junya_pattaraar@yahoo.com)
Yawarat Porapakkham (yawaratp@yahoo.com)
Alan D Lopez (a.lopez@sph.uq.edu.au)

Version: 5 Date: 3 May 2010

Author's response to reviews:

Manuscript revised as per email from PHM dt 29 April 2010. Reference number 10 removed; all remaining references and in text citations renumbered accordingly.